Author's response to reviews

Title: ALA-D activity is a reliable marker for oxidative stress in bone marrow transplantation patients

Authors:

Thissiane L Gonçalves (thissianegoncalves@yahoo.com.br)
Dalila M Benvegnú (dalilabenvegun@yahoo.com.br)
Gabriela Bonfanti (gabibonfanti@yahoo.com.br)
Andressa V Frediani (dessinhaf@yahoo.com.br)
João Batista T Rocha (jbtrocha@yahoo.com.br)

Version: 8 Date: 1 May 2009

Author's response to reviews:

I would like to do 2 changes in the manuscript.

First, reference 23 should be exchanged for a more appropriate one. The new one must


Second,

On the final of page 17 (lines 3-6, from the bottom to the top, the correct sentence is:

"In fact, an increase of ALA is typically associated with porphyria, hereditary tyrosine and lead poisoning and its accumulation causes oxidative stress and may lead to cancer [53]."

and not

"In fact, an increase of ALA is typically associated with porphyria, hereditary tyrosine and lead poisoning and its accumulation causes oxidative stress and may lead to cancer [53]."

because we are referring to poisoning with Pb (i.e., lead).

OK?